Pantograph-catenary interaction: recent achievements and future research challenges

S. Bruni^{1*}, G. Bucca¹, M. Carnevale¹, A. Collina¹, A. Facchinetti¹,

¹Dipartimento di Meccanica, Politecnico di Milano

Via La Masa 1

I-20156 Milano, Italy

Post-print version

*corresponding author: <u>stefano.bruni@.polimi.it</u>

To cite this paper:

Bruni, S., Bucca, G., Carnevale, M., Collina, A., Facchinetti, A. Pantograph–catenary interaction: recent achievements and future research challenges (2018) International Journal of Rail Transportation, 6 (2), pp. 57-82. DOI: 10.1080/23248378.2017.1400156

https://www.tandfonline.com/doi/full/10.1080/23248378.2017.1400156

ABSTRACT

This paper aims to provide an overview of the present status of research in pantograph-overhead line interaction and to outline future research challenges. A review of currently used modelling and simulation approaches is provided, also including hybrid simulation, and emerging trends are outlined. Then, line testing methods for the certification of pantographs and catenaries are described and the limitations of measuring methods now in use are highlighted. Issues related to the reliability and availability of the pantograph-overhead line system and to life cycle costs are then covered in the paper, particularly concerning wear and other damage phenomena and condition monitoring solutions enabling the efficient maintenance of the system. Pantograph aerodynamics and active control of pantographs are finally identified as two emerging trends of research. For both subjects, a review of problems currently at stake and of future research directions is provided.

KEYWORDS

Pantograph-catenary dynamic interaction, numerical simulation, HIL, laboratory and line tests, pantograph aerodynamics, active control

1 Introduction

In recent years, railways have known a substantial expansion across different continents. The impressive development of high-speed lines in China [1], the continuous expansion of the European network, the launching of direct train connection between China and Europe for freight and key projects launched in India, Turkey, former USSR Republics are just examples of the global tendency towards an increasing importance of railways in meeting transport needs for the global world. In this framework, improving pantograph-catenary interaction is a key issue towards enabling faster and more reliable operation of trains and, at the same time, reduce the maintenance costs for the rolling stock and the infrastructure.

For this reason, the effort spent in research on topics related to pantograph-catenary interaction has increased substantially over the last years: a search performed on the SCOPUS database shows that in years 2005-2006 only 19 papers were published having as their subject pantograph-catenary interaction. In years 2010-2011 this number had grown to 48 and in 2015-2016 there were 80 papers published on this subject.

The increased interest on pantograph-catenary interaction topics is also demonstrated by the funding of two research projects in the sixth and seventh framework programmes of the European Commission, namely EUROPAC [2] and PantoTRAIN [3], that led to interesting developments and helped to identify various challenges. Progresses in the establishment of mathematical / numerical models for the simulation of pantograph-catenary interaction were recently reviewed by a benchmark exercise that involved several research institutes across 9 countries in Europe and Asia [4]. The results of this benchmark allow to draw conclusions on the present status of simulation tools and also to provide a term of reference for new models being developed [5]. However, the benchmark strictly focussed on mathematical modelling and simulation, whilst there are other important areas in which research related to pantographs and catenaries is quickly developing.

It is therefore the aim of this paper to provide a comprehensive overview of recent research concerning all mechanical aspects related to the interaction between the pantograph and the overhead equipment (catenary). These include not only modelling and simulation, that are only briefly addressed having been the subjects of past [6] and recent [4,7] comprehensive works, but also testing and measurements (both in laboratories and in the line), wear and damage mechanisms, aerodynamics, health monitoring and active pantograph control. This review paper will not cover, however, aspects strictly related to the electric interaction between the pantograph and the catenary such as modelling of arching, electro-magnetic compatibility, electric disturbances generated by the train on the overhead power supply (so-called 'harmonics') and so on.

The paper is organised as follows: Section 2 provides an updated overview of the State-of-Art concerning the numerical simulation of pantograph-catenary interaction. Section 3 deals with the so-called 'hybrid' or 'Hardware-in-the-Loop' simulation of pantograph-catenary interaction, representing a recent development and offering prospective advantages over 'standard' numerical simulation. Section 4 describes methods for the testing and qualification of pantographs, with a focus on the measure of pantograph-catenary contact forces. Section 5 covers issues related with the condition monitoring and with wear and damage in the pantograph-catenary couple. Section 6 provides an updated State-of-Art review on the investigation of pantograph aerodynamics using both experimental and numerical methods. Section 7 summarises recent investigations concerning the possibility to improve current collection by means of active pantograph control. Finally, Section 8 draws some conclusions and identifies needs for further research.

2 Numerical simulation of pantograph-catenary interaction

Numerical simulation tools for the analysis of pantograph-catenary dynamic interaction are well established. Comparing the state of the art in mid 1990's [6] and in recent years [4,7] the evolution and spread of numerical models can be easily appreciated. Early approaches were based on simplified representation of the catenary as a spatially variable stiffness [8], whilst together with the increase of computational capabilities, a variety of simulation approaches has been developed and proposed. Most of the available tools are based on finite element modelling of the overhead

catenary line (OCL), as shown in Figure 1, and lumped mass pantograph models (Figure 2a), adopting a penalty method to describe the flexible mono-lateral contact between the catenary and the pantograph [9-16]. In [17] a more general multibody model (Figure 2b) is introduced for the pantograph: this model is run in co-simulation with a finite element catenary model. The finite difference method is considered as an alternative to the finite element method for the spatial discretization of the catenary in [18] and is adopted by PrOSA [19]. The finite element or the finite difference models are the best choice for modelling the catenary because they allow to reproduce the effects of transversal wave propagation along the wires, which become very important for the multi-pantograph case. The slackening of the droppers results in nonlinear effects strongly affecting the dynamics of the pantograph-catenary system and, therefore, the inclusion of these effects also represents a key requirement in order to achieve realistic simulation results ([10,13]).



Figure 1. Example of FE representation of an OCL section.



Figure 2. Pantograph models: a) lumped parameters b) multibody.

As demonstrated by the recent benchmark among simulation codes for pantograph-catenary dynamic interaction ([4,5]), affordable simulation tools are nowadays available and have reached a certain level of consensus within the scientific and technical international community. Most of these tools can be used for verification against normative requirements (e.g. the Technical Specifications for Interoperability – TSI in Europe) of the general design of a standard OCL, including overlap sections and insulators ([20,21]). Other OCL features sometimes need to be considered, such as neutral section with insulators, transitions to tunnels, rigid catenaries and the related transitions to and from standard OCLs: the possibility to model these special features of the line partially depends on the used approach. Under this respect, without discarding the importance of semi-analytical methods, the finite element method is by far the most versatile approach for modelling an OCL, especially when particular configurations are investigated.

Some numerical challenges and open points were pointed out in the "statement of methods" papers describing the numerical tools compared in the benchmark [4]. In particular:

 the initialisation of catenary models is a challenging task. In fact, the loaded geometry is reached after nonlinear deformations of the structure and, since most of the simulation tools adopt linear finite element models for the catenary, obtaining realistic initial geometry using a linear FEM code could be not trivial [22,23].

- a detailed analysis of actual catenary geometries and construction details, and of the kinematic compatibility between the pantograph bow and the contact wire may require the adoption of a full 3D representation of the overhead structure, taking properly into account also the presence of curves in the railway track [22,23].
- the adoption of multibody approaches can improve the representation of the pantograph, especially at design stage when it is not possible to experimentally identify/verify lumpedmass model. Solver requirements are significantly different for linear FE models and for large rotation multibody models and the best way to solve this inconsistency appear to be the adoption of co-simulation approaches [7,22,24,25].
- the structural damping associated to catenary elements appear to be a key parameter for accurate representation of the pantograph-catenary interaction, in particular when dealing with multiple pantograph operation. The improvement of simulation models is related, among others, to a better identification [26] and modelling of catenary damping, as pointed out in [7].

In the last years, researches have mainly focused on the development of efficient numerical method for real-time execution of catenary models (see next section) and to the generalisation of catenary models to large deformation analysis [27-31], adopting absolute nodal coordinates [27-29] or corotational formulations [30]. This aspect is of particular importance for studies concerning crosswind effects on OCL [30,31].

As far as the subject of simulation is concerned, due to the recent request to upgrade rigid catenary systems for speed up to 200 km/h and more, the need to create numerical model for this kind of catenary system is emerging for design aims [32-35]. In this case, it is mandatory to include all the features related to tolerances in installation, misalignment in correspondence of bolted joints, deviation from exact rectilinear shape of the standard bars, and other structural features such as transitions with the standard wires catenary, insulating sections and switches.

One additional challenge that can be tackled by means of numerical simulation is the optimisation of catenaries and pantographs for interoperability. It has been shown that the enhancement of catenary systems can greatly benefit from using optimal approaches to allow train operations at higher speeds in the existing networks [36] and that the national pantograph-catenary pairs are basically optimised with respect to the interaction quality [37]. The optimisation of pantograph for cross network operation is however a still open challenge.

3 Hybrid simulation of pantograph-catenary interaction

Hybrid simulation has recently emerged as an enhanced way of simulating the dynamic behaviour of complex systems. The idea underlying hybrid simulation is to combine a physical hardware, representing one part of the dynamic system of interest, with a numerical model representing the remaining part of the system. The interaction between the hardware and the numerical model is established by a suitable testing apparatus and a real-time computing facility.

Hybrid simulation can be applied to the study of pantograph-catenary interaction considering the pantograph as the hardware part of the system which is interfaced to a numerical model of the flexible catenary: as shown in Figure 3, the force generated by the physical pantograph at the contact strips is measured (blue line) and is used to compute in real time the dynamic displacement of the contact wire at the points of contact with the pantograph according to a numerical model of the flexible catenary; the displacement (red line) is applied on the pantograph head by one or more actuators (in black in the figure).

The advantage with respect to pure numerical simulation is that modelling errors related with the modelling of the pantograph can be avoided. These errors are typically caused by neglecting or simplifying non-linear effects (due to e.g. friction, backlashes, kinematic nonlinearities) and flexibility effects in some pantograph components (frame, contact strips). At the same time, when compared to line testing, hybrid simulation is much less demanding in terms of the experimental work entailed, enables the investigation of innovative solutions such as active pantograph control without the risk of causing harm to the railway infrastructure and allows to perform investigations under repeatable conditions e.g. eliminating the effect of environmental factors.



Figure 3. The concept of hybrid simulation of pantograph-catenary interaction

Hybrid simulation of pantograph-catenary interaction was originally proposed by Zhang et al. at the State Key Laboratory of Traction Power of Southwest Jiaotong University [38,39], using a linear model of the catenary based on modal superimposition. A refinement of this approach was provided by Resta et al. [40] by considering the effect of dropper slackening in the real-time model of the catenary. Facchinetti and Bruni [41] further refined the method by introducing the independent actuation of the two collector strips and by considering the effect of the stagger in the contact wire. Also in [41] the accuracy of the hybrid simulation approach is for the first time assessed by means of extensive comparison with line test measurements. The use of hybrid simulation for the design of overhead lines for very high speed (360 km/h and above) is discussed in [42] and compared with an approach based on 'pure' numerical simulation.

One main line of development for hybrid simulation deals with improving the control strategies of the testing apparatus realising the interaction between the physical pantograph and the virtual catenary. In [43] Stoten, Yamaguchi and Yamashita propose the Dynamically Substructured

System as an alternative to standard hybrid simulation, providing enhanced stability and robustness against uncertainties in system's parameters The method is applied to a so-called 'quasi-pantograph' i.e. a mass-spring system whereas the catenary is considered as a linear single degree of freedom system with time varying stiffness. In [44] Schirrer et al. propose an integrated impedance control strategy capable of compensating for the phase lag introduced by the control unit of the testing apparatus. The synthesis of this controller is formulated in terms of a linear model predictive control and hence is, in strict sense, not compatible with considering the non-linearities implied by dropper slackening and contact loss between the pantograph and the contact wire.

Another subject of present research is concerned with improving the accuracy of real-time models of the catenary. Facchinetti, Gasparetto and Bruni in [45] considered a class of models based on modal superposition and assessed the effect of changing the number of modelled spans and the number of modal components used to describe the vibration of the contact and messenger wires in view of finding the best model compatible with real-time simulation. In [44] an Eulerian model of the catenary, i.e. described in pantograph-fixed coordinates, is proposed. The model includes boundary absorbing layers to reduce wave reflection of waves at its boundaries. In this way, the computational effort required to solve the model is kept small enough to enable real-time computing and at the same time the motion of the catenary can be described to a good degree of detail. This approach however neglects dropper slackening, which is a major effect in pantographcatenary dynamics. In S. Gregori et al. in [46] the numerical integration of the catenary equations of motion is decomposed in an "offline stage" and an "online stage". In the offline stage the time-step response of the contact wire and messenger wire to unit loads applied at discrete positions along the contact wire and at the dropper ends are calculated and stored. In the online stage the pre-calculated responses are used to reduce the number of dynamic equations to one for each pantograph plus one for each dropper, resulting in a computational effort compatible with real-time simulation. This method fully accounts for dropper slackening, whereas the catenary is modelled using the Finite Element method according to the Absolute Nodal Coordinate Formulation, thus allowing to represent in detail the geometry of the catenary, including the actual non-rectilinear shape of the messenger wire. Although this method has not been used yet for hybrid simulation, it looks as being well suited to provide a good balance between accuracy and computational effort.

4 Testing and qualification of pantographs and ocl

4.1 Generalities

The qualification (or homologation) process of a pantograph or of an OCL is a complex process, involving technical and managing issues and composed of several phases, from design to laboratory testing of the pantograph, simulation of pantograph-OCL dynamics in order to demonstrate the feasibility of the operation, and finally the testing of the pantograph on the rolling stock, in a series of trial tests. The aim of qualification is to ensure that the operation of a pantograph or of a OCL will take place regularly and without negative effects on OCL and rolling stock maintenance.

Qualification may take place at national level, applying national rules and standards, or at international level, e.g. applying TSI specifications [47] recalling proper EN/IEC standards [48-50]. The aim of the TSI process, strictly valid for the European network, is to assess the capability of a TSI-compliant pantograph to run under different TSI-compliant OCLs, and vice versa. In both cases the assessment involves mechanical and electrical issues. Considering line tests, with particular reference to qualification process, the following parameters are usually taken into account:

mean contact force exchanged between pantograph and OCL as a function of train speed, this is related to the uplift of the OCL (especially under suspension, see point below). It is worth mentioning that current TSI provide a limit to the mean contact force up to 320 km/h. For higher speeds, national rules are applied or need to be established. In a recent qualification, the possibility to operate the pantograph setting a constant upper threshold for the mean contact force, from 320 km/h to 360 km/h, was analysed through simulations and verified with line tests [51];

- OCL uplift at suspensions, this is an exclusive mechanical issue related to the safety margin with respect to the engagement of the stagger arm by the collector that is passing by;
- dynamic component of the contact force exchanged between the collectors and the contact
 wire. This has to do with the mechanical and electrical issues involved in the electrical
 power transfer from the OCL to the drives of rolling stock's motors. Too high contact force
 may induce local mechanical damage on the OCL, whereas excessive occurrence of low
 values of contact force correspond to continuous sparking and arcing that causes electricalrelated wear on collectors and contact wires, and disturbance to the signalling system or to
 the drives of the motor;
- percentage of contact loss, that represents the occurrence of arcing revealed during the pantograph collecting periods. At TSI level the contact force dynamic variation and the percentage of contact loss measurement are addressed as alternative quantities for the assessment of current collection.

4.2 Contact force measurement

The contact force between pantograph and catenary is directly related to the quality of the current collection. It is usually indirectly measured as the sum of the following three terms [52-54]:

- the forces acting between each collector and its suspension, measured by load cells mounted between each collector and its suspensions (see Figure 4);
- a term compensating the inertial effects in the collector strip, measured by two accelerometers located close to the load cells (Figure 4);
- a term compensating for the mean aerodynamic lift forces acting on the collector strips. This
 last contribution becomes important only above 150 km/h and is obtained by means of line
 tests performed on a restrained instrumented pantograph (Figure 5) in a wind tunnel or in the
 line.

The force measuring system described above is not free from drawbacks: the frequency band of the measure is limited to 20 Hz, since the compensation of inertial effects considers the collector as a rigid body. Moreover, the force transducers become part of the structure and must therefore ensure sufficient mechanical strength, generally conflicting with the sensitiveness required for the range 0.500N typical of this application, and the impact of the transducers to the behaviour of the pantograph in terms of mass and aerodynamic effects is sometimes not fully negligible.



Figure 4. Sketch of the contact force measurement setup.



Figure 5. Example of instrumented pantograph for contact force measurement.

An alternative measurement setup is the one with strain gauges directly installed on the connection between the collector strip and the frame of the collector, as reported in [55], considering the contact force as proportional to the strain in such locations.

All the above described measurement setups are based on sensors requiring electric power supply: for the pantograph-catenary application this is problematic given that the pantograph works under high voltage. To guarantee the safety of the whole system, the most adopted solution features the use of a pack of batteries safely secured on the roof of the vehicle, nearby the pantograph. In addition, a second problem can arise: in order to ensure full electrical insulation, all the electrical signals outputted by the sensors must be converted into optical signals by means of converters and transferred to the data acquisition system (usually inside the vehicle) using fibre-optic cables. These signals then can be acquired directly or converted back from optical to electrical signals with other converters and sent to the acquisition board ([56]). To avoid problems related to the power supply of sensors and to the electrical insulation, in [51] and [57,58] optical load cells and optical accelerometers are proposed instead of electrical sensors. Optical sensors benefit from being intrinsically electrically insulated and insensitive to electromagnetic disturbances. Moreover, they do not need any electric power supply. These features make the optical sensors very suitable for the use on a pantograph and make the measurement setup very easy and with very low impact with respect to a traditional measurement setup.

In some application, as reported in [59,60], in order to simplify the measure setup, fibre optic sensors are directly connected to the collector strip and used to measure strains due to the interaction between strip and wire. This relatively simple and robust measuring set-up can be installed on pantographs used in commercial service and hence used for monitoring aims, see also Section 5.

The future developments of set-ups for contact force measurement are addressed to overcome the frequency limit of 20 Hz [61]. In any case, to this aim it is mandatory to consider collector deformability [62].

4.3 Contact loss detection

The standard method for the contact force measurement unavoidably modifies the original geometry and mass of the pantograph collector and its suspension, even if the measuring set-up is designed paying particular care to produce the minimum impact on the pantograph. An alternative method to evaluate the current collection quality is to detect contact losses through the observation of electrical arc generation. As reported in [63,64], the method is not invasive and is based on the optical measurements of the duration of ultraviolet emission due to electric arcing. The optical method can be used for both a.c. and d.c. current systems, while just for d.c. current system a method to detect contact loss based on the measurement and analysis of pantograph voltage drop is proposed in [52].

4.4 OCL motion

The analysis of pantograph-catenary interaction is generally completed by measuring catenary motion. In particular, the analysis of contact wire vertical uplift under the suspension (Figure 6) allows to verify the effect of pantograph load on the wire and to assess the safe and correct interaction between the pantograph and the catenary.

In [65] a dedicated wireless device is developed and proposed to measure wire accelerations at any location of interest, enabling to analyse/assess the motion of the OCL.



Figure 6. Example of contact wire uplift measurement.

5 Damage, condition monitoring and fault detection

Regularity of traffic operation is nowadays a highly mandatory issue for any railway network at international, national, regional or local level. The effect of traffic disruption can be costly in terms of lost hours of operation and impact on people's mobility. From this point of view, rolling stock and infrastructure maintenance is a key issue for maintaining the required regularity of train operation.

In the case of the pantograph -OCL couple, a failure of one of the two subsystems may have in the worst case a catastrophic consequence on both, the failure on the OCL being much more impacting on traffic disruption.





b) Figure 7 a) worn srface of a contact wire; b) worn thickness of a carbon contact strip.



Figure 8 Picture and schematic views of the test bench developed at PoliMi ([71,75-77]).

One aspect of maintenance is related to wear phenomena which affect pantograph collectors and the contact wire (Figure 7). Researches in this field are aimed at the reduction of wear levels on both collectors and wires, to decrease life cycle costs. Some test benches (Figure 8) were developed to reproduce and to study the combined electro-mechanical phenomena occurring at the contact between collector strip and contact wire. Current intensity, mean contact force and arcing are generally recognized as mutually influencing factors in most studies performed. In particular, the reciprocal influence of mechanical and electrical wear is discussed in [66,67], while the effect of arcing on the wear of collectors and wires is analysed in [68,69], introducing the electrical discharge energy concept, and in [70,71] considering a.c. current collection. The influence of different types of contact wire and collector strip material on the wear is investigated by different researchers: pure carbon contact strip in [72], copper impregnated carbon strip in [68], iron based strip in [69], special sintered materials in [73] and reinforced carbon composites in [74] are just some examples. Finally, a procedure to extend laboratory data to operating conditions is proposed in [75-77]. In these papers, a numerical model for surface wear is introduced and used in combination with a method for the numerical simulation of pantograph-OCL interaction to estimate the wear rate of contact strip and contact wire.

The continuous increase of operation speed led some researchers to investigate new aspects related to OCL damage, as in [78,79] where fatigue problems in the contact wire are discussed.

To evaluate the correct actions to maintain the OCL, several means and approaches are available:

- visual and manual inspection (f.i. contact wire thickness verification), performed on regular time base. This activity is carried out mostly at local level, and its effectiveness is somehow depending on the skill and organisation of the local maintenance crew;
- measurement of contact wire (CW) geometry (vertical and lateral position) and thickness, using special inspection vehicles or trains. In this second case, rather sophisticated transducers are used to measure directly the geometrical parameters of the contact wire, in relationship with the location along the line. Depending on the class of the line (local, regional, high speed) the inspection is carried out at time intervals ranging from two months up to one year. Such systems are run by a skilled crew who manages the overall measurement process, and the results are analysed by specialised teams belonging to the maintenance management structure of the infrastructure manager;
- a new emerging option is to use commercial trains equipped with simple transducers (typically accelerometers) and to use the dynamic response of the pantograph as an indicator to detect defects along the line, or to assess the general OCL's status in terms of contact wire vertical irregularity [80]. The use of commercial trains allows a daily data flow, and is the

prerequisite for the development of Condition Based Monitoring techniques, off-line trend analysis and Condition Based Maintenance.

In CBM systems the acceleration of the collector is typically used [81-83], since it is highly related to the contact force [81], both in terms of peak values and standard deviation. The issue of electrical insulation of sensors, together with the additional problem of using an easy-to-maintain measurement set-up, comes into play to develop a system suitable for installation on commercial-service trains. Optical accelerometers were especially developed for this application [81]. The loss of information inherent in the use of a set-up measuring only accelerations, instead of one also including the contact force measurement can be compensated by the larger amount of data acquired from several trains running under the same line. A cross-reference analysis of the data from different trains allows strengthening the identification of local and distributed defect of the OCL. Moreover, the comparison of different trains (and hence pantographs) on the same line enables to isolate those pantographs that are not in standard maintenance condition, so excluding their data for the analysis, and, at the same time, lead them to maintenance to restore their standard condition.

One main difference between CBM approach and the standard approach based on specialised inspection train is the far larger amount of acquired data, in terms of frequency of inspections and number of trains that acquire such data. If on one hand the availability of more data allows performing a more robust analysis, on the other hand it poses problems in terms of data management. To overcome this problem, a real-time processing of the data on-board train is generally proposed. Suitable algorithms allow reducing the large amount of raw data to a manageable, but still significant, set of parameters representing the status of the line [81].

To drive maintenance activities, the above mentioned CBM indexes (synthetic parameters) must be related to time stamp, GPS coordinates, or milestone position along the line. The latter can be evaluated on-board train when performing real-time analysis, and on-going research and experimental activities are nowadays aimed at improving the accuracy of algorithms and hardware for train positioning. This set of data must then be stored in a way side server, which enables further comparison with previously acquired data (trend analysis), to localise the defect, and to set the date of a damage event. The comparison between daily monitored situation and reference data will allow enhancing the detection of defects and the analysis of their long-term development, as well as to reveal sudden changes in the status of the line and take proper countermeasures before a serious failure occurs.

Finally, it should be mentioned that also other techniques have been proposed for the monitoring of the pantograph-catenary system, such as infrared camera to detect the temperature along the strip [84], and CCD and digital cameras [85,86] to infer relevant aspects regarding the status of pantograph and overhead line condition.

6 Experimental and computational aerodynamics

Pantograph and overhead line aerodynamics is responsible for affecting the contact force between the pantograph and the contact wire, both in terms of average value and dynamic variation [87-89]. This section deals with aerodynamic forces acting on the pantograph, the issue of wind effects on catenary having already been addressed in Section 2. When evaluating the pantograph performances for train speed higher than 200 km/h, aerodynamic forces on pantograph need to be considered, both when performing numerical simulations or when adopting experimental approaches.

The major aerodynamic effect is related to mean drag and lift forces acting on the moving parts of the pantograph, which affect the mean value of the contact force adding their contribution to the uplift force exerted by the pantograph raising mechanism at the bottom of the articulated frame (normally an air spring) [88,90]. This effect, usually addressed to as *aerodynamic uplift*, is dependent on train speed, pantograph working height [91] and orientation [92,]. Modern pantographs have indeed an asymmetrical geometry generating different aerodynamic uplifts in the two orientations in which they can operate. Moreover, the aerodynamic uplift varies when the pantograph enters a tunnel, due to the increase of the velocity of the relative flow [94]. Attempts to balance the aerodynamic uplift by means of aerodynamic spoilers were made in order to guarantee

operational stability, but this was not trivial considering that it is very difficult to optimise the spoilers for both pantograph orientations. Therefore, in recent years, the regulation of air-spring pressure as a function of train speed and pantograph orientation has been proposed as a means to compensate for the aerodynamic uplift and to guarantee the best performances in both running directions and at all speeds [15].

When pantographs with two independent collectors are adopted, aerodynamic steady forces can also generate an unbalance between the mean contact forces of the front and rear collector [87,92], with a negative outcome on power collection and a non-uniform wear of the two strips.

Aerodynamic non-stationary phenomena also influence the performance of a railway pantograph, and can be divided in two groups, related to the turbulence of the incoming flow and to vortex shedding. The presence of recesses, coach separation, electrical insulators, switches and other components installed on the train roof generates a turbulence wake, whose frequency spectrum is likely to excite the pantograph structure also within the frequency range set by international standards for the evaluation of the quality of current collection (0-20 Hz in Europe) [50,87,89]. In some cases, the turbulence can be even generated by pantograph components, such as the upwind collector or the air spring placed at the bottom frame, with the consequence that the rear parts of the pantograph are immersed in the turbulent wake.

Vortex shedding occurring in the wake of pan-head and collectors [96] should also be considered as a relevant issue, since it introduces very high-frequency excitation which not only has an impact on aerodynamic noise [97], but can also worsen the current collection quality, by exciting collector flexural modes of vibration and increasing the level of arcing [62].

Many research works have investigated the possibility to reduce the impact of aerodynamic effects on the quality of current collection. Several of them consider the problems due to average forces [95,98,99] and few consider non-stationary phenomena [100], mainly in relation to aero-acoustic issues [99,101]. At present, only average effects are regarded when evaluating the contact force variability, both in technical and scientific communities, as well as in EU international

standards. The topic related to the frequency content of the incoming wind turbulence and its interaction with the pantograph structure is not much treated in the literature, either from experimental or numerical point of view.

Experimental on-track tests are still the main instrument not only for the evaluation of pantograph aerodynamic performance during the homologation process, but also for the fine-tuning of the best design solutions. Wind tunnel experimental tests (Figure 9a) are a powerful tool, since they allow to assess pantograph aerodynamic behaviour for several test configurations, architectures and design solutions under controlled flow conditions, but their drawback consists in the difficulty of reproducing the actual boundary layer of the train roof, and the turbulence conditions of the wind flow. In [102] the Authors point out the importance of the boundary layer reproduction in determining the uplift force in wind tunnel. They propose an effective technique to develop a boundary layer with similar characteristics to the one preliminary measured on the real train, which needs anyway, again, on-track tests to be tuned and validated.



Figure 9 a) full-scale pantograph under test in a wind tunnel; b) example of CFD results in terms of steady state pressure over pantograph elements.

Computational fluid dynamics (CFD) does not yet represent a mature field for the evaluation of aerodynamic effects on pantograph contact force. CFD simulations have been performed in literature mainly focusing on the pantograph head and collectors model, in order to study drag and lift forces on these components [87] and acoustic emission [97,98,101,103]. Regarding the possibility of estimating aerodynamic forces on the entire pantograph, some authors have developed CFD models of a full-scale pantograph in a domain representing only the part of the carbody roof close to the pantograph [104], or CFD models of a pantograph installed on a full-scale train [105,106]. In [107], a full-scale pantograph is tested in a wind tunnel and the experimental results are compared with those of CFD models (Figure 9b). However, to the authors' knowledge, a complete validation of the CFD model against experimental results on a full-scale pantograph in wind tunnel and on a full-scale train became available only in the very last years in [92,93], in which a procedure for the numerical evaluation of the aerodynamic uplift is developed, based on CFD results and the application of the virtual work principle.

7 Active pantographs

Active control of pantographs has been considered as a possibility to improve current collection quality for a long time [108-109], especially for high speed applications, with the aim of reducing contact losses and contact force variability. The adoption of active control appears in principle promising to solve interoperability issues, considering that the same pantograph may interact with different catenaries and a comprehensive optimisation of the interacting passive systems is thus not feasible.

Several studies can be found in the literature, but most of them are limited to the proposal of control strategies and to the theoretical or numerical verification of the achievable performances. Only in few documented cases, these studies came up to the realization of active pantograph prototypes, verified through laboratory experiments [110-113] or in line tests [114].

The proposed solutions can be classified on the basis of the actuation principle or of the adopted control strategies. Considering the actuation principles, the actuators for applying the control action are usually considered to operate in parallel to the passive system rather than to completely replace pantograph components, in order to keep the possibility to operate in case of failure of the control system. The control action can be applied on the articulated frame (Figure 10a), in parallel with the collector head suspension (Figure 10b) between the collectors and the articulated frame, or directly on the collectors (Figure 10c).



Figure 10: Actuation principles for pantograph active control: a) control action on the articulated frame; b) control action in parallel with the head suspension; c) control action applied directly on the collectors.

The first configuration is the simplest and has the advantage that there are no strict limitations to the size, power, mass and kind of actuator (pneumatic, hydraulic or electric), since the actuator can be placed directly on the vehicle roof. The pneumatic bellows equipping the pantograph for the raise/lower function and for providing a constant preload can be directly regulated by pneumatic valves to achieve a simple preload regulation, as done on the TGV fleet in France [115] or on the ETR500 and ETR1000 fleets in Italy [15], or to control the slow dynamics of the articulated frame [110,114,116,117]. As possible alternatives the introduction of an electric torque motor between the lower arm of the articulated frame and the vehicle roof is proposed in [118], the use of an hydraulic actuator in parallel to the pneumatic bellows of the articulated frame is reported in [6] and also adopted on a prototype in [113], whereas a wire actuation system connecting the extremity of the upper arm and actuated by an electric motor mounted on the vehicle roof is used in [111]. The main drawback lies in the limited bandwidth that can be achieved (up to **1-2Hz**), on account of the filtering action introduced by the head suspension [119,120].

The second configuration, with the control action in parallel to the head suspension, is more effective for the control of the medium frequency range (up to 20Hz) but imposes limitations to the size and mass of the actuators that are placed in correspondence of the collector head suspension,

thus increasing the overall mass of the pantograph. The possible implementations strongly depend on the solution adopted for the head suspension. The possibility to adopt a hydraulic actuator in parallel to the head suspension is reported in [6]. The adoption of a linear electric motor in parallel to the linear springs of the head suspension is considered in [118], while a torque motor is proposed in [120] on account of the different configuration of the head suspension. In [110,114] the head suspension consists of two torsion bar springs and the active control is introduced using two pneumatic actuators placed beside the two torsion bar springs.

The last configuration, with the control action directly applied on the collectors, is the most difficult to implement. Yet some possible solutions have been proposed, like the adoption of a wire actuation system connecting electric motors on the roof to the collectors [111,121,122] or the use of a movable wing placed just below the collector [123], driven by an electric motor and introducing a control action in the form of aerodynamic lift force.

Obviously, the different actuation configurations can be adopted at the same time, as described in [6,110,114].

As far as control strategies for the active pantograph are concerned, the proposed solutions can be divided in two macro-categories, i.e. those that consist in controlling directly the contact force and those where active control is used as a means to modify and improve the dynamic behaviour of the pantograph.

As far as contact force control is concerned both classical control techniques and modern control techniques were proposed. Contact force feedback with a PI regulator is investigated and its effectiveness proved in [121] considering a simplified model of the catenary and in [120] considering a detailed mathematical model of the pantograph-catenary interaction and of the electric actuation. PID adoption is considered in [124,125] and in [126], with the addition of a notch filter and particle swarm optimisation technique. In [110] the control system is made of two stages, one on the pneumatic bellows for slow dynamics and the other in parallel to the head suspension, both characterised by classical controllers. Considering modern control techniques, many different

approaches can be found in literature, including but not limited to sliding mode control [122] and fuzzy logic [127,128].

It is worth remarking that all these strategies require either the direct measure of the contact forces (using the measuring set-ups described in Section 4) or an estimate of this quantity. In this respect, the adoption of an Extended Kalman Filter is proposed in [129] and verified with experimental data in [130]. An algebraic observability approach is considered in [131], while a sliding mode observer is proposed in [132].

A possibility to overcome the need for contact force knowledge is to use a different strategy for active control, aimed at modifying the dynamic behaviour of the pantograph. In this regard, optimal control has been considered by different authors [133-136] as a means to improve pantograph response. As an alternative, an approach based on the concept of mechanical impedance simulation is propose in [113], where the control action is used to emulate the behaviour of a desired mechanical system, which is not realizable using only passive components. In [137] sliding mode control is used to improve the mechanical impedance of the pantograph head and the approach was verified considering also the presence stochastic wind.

A direct comparison of the surveyed control strategies and of their effectiveness is extremely difficult, since they were applied to different pantographs and their evaluation was performed by means of different numerical models, characterised by different complexity levels (e.g. considering or not the control actuator dynamics).

As already mentioned, only a few proposed active systems have reached the prototype stage, so that further research is needed in order to verify the possibility to actually implement the suggested control strategies and to demonstrate their actual effectiveness in improving current collection quality.

8 Conclusions

This paper presented an overview on issues related to pantograph-catenary dynamical interaction. These can be summarised in the remarks listed below.

- (1.) The numerical simulation of pantograph-catenary dynamic interaction is nowadays a mature approach for the design and prequalification of an overhead equipment. In order to be effective for the entire design of a OCL from the mechanical and dynamical point of view, also the singular sections and transitions need to be considered in the model. Future developments in this field are represented by the analysis of environmental effects (e.g. wind, temperature) on the motion of the OCL and hence on its interaction with the pantograph.
- (2.) Despite being a relatively new approach to the study of pantograph-catenary interaction, hybrid simulation has been shown already to provide a degree of accuracy comparable to the one of 'pure' numerical simulation. Further important enhances to this technique are expected from the development of efficient methods for the real-time numerical simulation of complex catenary models and of course from the improvement of real-time computing hardware performances.
- (3.) The simulation of pantograph-catenary dynamic interaction coupled with models of wear and fatigue in the contact wire can be used to establish reasonable parameters for the evaluation of life cycle costs of the OCL. Similar methods can also be used to optimise costs related with the maintenance and replacement of collector strips.
- (4.) The improvement of methods for measuring the pantograph OCL contact force is one important challenge related with the testing and qualification of new pantographs and catenaries. Future research needs in this area are concerned with extending the frequency range of the measure beyond the present upper limit of 20 Hz. Furthermore, the use of innovative sensors based on fibre-optics is a very promising field of research.
- (5.) Line measurements of pantograph / OCL vibration and of the contact force are also relevant to the continuous monitoring of the OCL and with the implementation of Condition-Based Maintenance. In this regard, present and future research topics are concerned with developing robust and cost-effective measuring set-ups and with the analysis of measured

data. The use for continuous monitoring of train fleets operated in commercial service opens new scenarios but requires the use of suitable 'big data' techniques.

- (6.) Pantograph aerodynamics is extremely important in very-high speed applications. At present, the aerodynamic optimisation of pantographs is mostly carried out by means of experiments. It is possible that in the future some part of this development will be undertaken using computational fluid dynamics methods, but this requires a further enhancement of CFD techniques.
- (7.) Finally, active pantograph control can be used as a means to improve pantograph-catenary interaction, e.g. operating at higher speeds pantographs under existing catenary that are possibly not optimised for very high speeds. However, there are few examples of demonstration of this technology up to the prototype stage and the effectiveness of active systems in improving current collection quality in real operation hasn't been yet demonstrated. Research challenges in this area are concerned with developing actuation methods and active control strategies.

Acknowledgments

The Authors would like to gratefully acknowledge the "Joint Research Centre on Railway Transports" established by Fondazione Politecnico di Milano for providing financial support to this work.

References

- [1] Zhou L, Shen Z. Progress in high-speed train technology around the world. J Mod Transp. 2011;19(1):1-6. doi: 10.3969/j.issn.2095-087X.2011.01.001.
- [2] Bobillot A, Cléon LM, Collina A, Mohamed O, Ghidorzi R. Pantograph-catenary: a highspeed European couple. 8th World congress on railway research (WCRR 2008); 2008 May 18-22; Seoul, Korea.

- [3] Hanley R, Liciardello L, Martin S, Rose K. TrioTRAIN: bringing research and standardisation closer together. TRA2014 Transport Research Arena 2014, Transport Solutions: from Research to Deployment - Innovate Mobility, Mobilise Innovation!; 2014 Apr 14-17; Paris, France.
- [4] Facchinetti A, Bruni S. Special issue on the pantograph–catenary interaction benchmark. Veh Syst Dyn. 2015;53(3):303-304. doi: 10.1080/00423114.2015.1017308.
- Bruni S, Ambrosio J, Carnicero A, Cho YH, Finner L, Ikeda M, Kwon SY, Massat JP, Stichel S, Tur M, Zhang W. The results of the pantograph–catenary interaction benchmark. Veh Syst Dyn. 2015;53(3):412-435. doi: 10.1080/00423114.2014.953183.
- [6] Poetsch G, Evans J, Meisinger R, Kortüm W, Krabacher W. Pantograph/catenary dynamics and control. Veh Syst Dyn. 1997:28(2-3):159-195. doi: 10.1080/00423119708969353.
- [7] Ambrosio J, Pombo J, Pereira M, Antunes P, Mósca A. A computational procedure for the dynamic analysis of the catenary-pantograph interaction in high-speed trains. J Theor Appl Mech. 2012;50(3):681-699.
- [8] Wu TX, Brennan MJ. Dynamic stiffness of a railway overhead wire system and its effect on pantograph-catenary system dynamics. J Sound Vib. 1999;219(3):483–502. doi: 10.1006/jsvi.1998.1869.
- [9] Fujii Y, Manabe K. Computer simulation for dynamics of overhead catenary-pantograph system. 3rd International Conference on Computer-Aided Design, Manufacture and Operation in the Railway and Other Mass Transit Systems (COMPRAIL 92); 1992 Aug 18--20; Washington DC, USA.
- [10] Collina A, Bruni S. Numerical simulation of pantograph-overhead equipment interaction. Veh Syst Dyn. 2002;38(4):261–291. doi: 10.1076/vesd.38.4.261.8286.
- [11] Massat JP, JP Lainé JP, Bobillot A. Pantograph-catenary dynamics simulation. Veh Syst Dyn. 2006;44(S1):551-559. doi: 10.1080/00423110600875443.

- [12] Zhang WH, Liu Y, Mei GM. Evaluation of the coupled dynamical response of a pantograph– catenary system contact force and stresses. Veh Syst Dyn. 2006;44(8):645–658. doi: 10.1080/00423110600744656.
- [13] Cho YH. Numerical simulation of the dynamic responses of railway overhead contact lines to a moving pantograph, considering a nonlinear dropper. J Sound Vib. 2008;315(3):433–454. doi: 10.1016/j.jsv.2008.02.024.
- [14] Lopez-Garcia O, Carnicero A, Maroño JL. Influence of stiffness and contact modelling on catenary-pantograph system dynamics. J Sound Vib. 2007;299(4–5):806–821. doi: 10.1016/j.jsv.2006.07.018.
- Bucca G, Carnevale M, Collina A, Facchinetti A, Drugge L, Jönsson PA, Stichel S. Adoption of different pantographs' preloads to improve multiple collection and speed up existing lines. Veh Syst Dyn. 2012:50(S1):403-418. doi: 10.1080/00423114.2012.665165
- [16] Tur M, García E, Baeza L, Fuenmayor FJ. A 3D absolute nodal coordinate finite element model to compute the initial configuration of a railway catenary. Eng Struct.2014;71:234–243. doi: 10.1016/j.engstruct.2014.04.015.
- [17] Ambrósio J, Pombo J, Rauter F, Pereira M. A memory based communication in the cosimulation of multibody and finite element codes for pantograph-catenary interaction simulation. Comput Methods Appl Sci. 2009;12:231–252.
- [18] Arnold M, Simeon B. Pantograph and catenary dynamics: a benchmark problem and its numerical solution. Appl Numer Math. 2000;34(4):345–362. doi: 10.1016/S0168-9274(99)00038-0.
- [19] Finner L, Poetsch G, Sarnes B, Kolbe M. Program for catenary-pantograph analysis, PrOSA statement of methods and validation according EN 50318. Veh Syst Dyn. 2015;53(3):305-313. doi: 10.1080/00423114.2014.958501.

- [20] Hare`ll P, Drugge L, Reijm M. Study of critical sections in catenary systems during multiple pantograph operation. Proc Inst Mech Eng F, J Rail Rapid Transit. 2005;219(4): 203–211. doi: 10.1243/095440905X8934.
- [21] Benet J, Alberto A, Arias E, Rojo T. A mathematical model of the pantograph-catenary dynamic interaction with several contact wires. Int J Appl Math 2007;37(2):136–144.
- [22] Ambrósio J, Pombo J, Antunes P, Pereira M. PantoCat statement of method, Veh Syst Dyn, 2015;53(3):314-328. doi: 10.1080/00423114.2014.969283.
- [23] Tur M, Baeza L, Fuenmayor FJ, García E. PACDIN statement of methods. Veh Syst Dyn, 2015;53(3):402-411, 2015. doi: 10.1080/00423114.2014.963126.
- [24] Massat JP, Balmes E, Bianchi JP, Van Kalsbeek G. OSCAR statement of methods. Veh Syst Dyn. 2015;53(3):370-379. doi: 10.1080/00423114.2015.1005016.
- [25] Ikeda M. 'Gasen-do FE' statement of methods. Veh Syst Dyn. 2015;53(3):357-369. doi:
 10.1080/00423114.2014.968174.
- [26] Nåvik P, Rønnquist A, Stichel S. Identification of system damping in railway catenary wire systems from full-scale measurements. Eng Struct. 2016;113: 71-78. doi: 10.1016/j.engstruct.2016.01.031.
- [27] Seo JH, Sugiyama H, Shabana AA. Three-dimensional large deformation analysis of the multibody pantograph/catenary systems. Nonlinear Dyn. 2005;45(1-2):109–130. doi: 10.1007/s11071-005-2789-9.
- [28] Seo JH, Kim SW, Jung IH, Park TW, Mok JY, Kim YG, Chai JB. Dynamic analysis of a pantograph-catenary system using absolute nodal coordinates. Veh Syst Dyn. 2006;44(8):615-630. doi: 10.1080/00423110500373721.
- [29] Kulkarni S, Pappalardo CM, Shabana AA. Pantograph/catenary contact formulations. J Vib Acoust Trans ASME. 2017;139(1):011010. doi: 10.1115/1.4035132.

- [30] Sanchez-Rebollo C. Wind effects on the dynamic interaction between pantograph and overhead contact lines [dissertation]. Madrid: Universidad Pontificia Comillas ICAI-ICADE; 2014.
- [31] Song Y, Liu Z, Wang H, Lu X, Zhang J. Nonlinear analysis of wind-induced vibration of high-speed railway catenary and its influence on pantograph-catenary interaction. Veh Syst Dyn. 2016;54(6):723-747. doi: 10.1080/00423114.2016.1156134.
- [32] Bautista A, Montesinos J, Pintado P. Dynamic interaction between pantograph and rigid overhead lines using a coupled FEM-multibody procedure. Mech Mach Theory. 2016;97:100-111. doi: 10.1016/j.mechmachtheory.2015.10.009.
- [33] Cho YH. SPOPS statement of methods. Veh Syst Dyn. 2015;53(3):329-340. doi:
 10.1080/00423114.2014.953182 SPOPS.
- [34] Collina A, Bruni S, Facchinetti A, Zuin A. PCaDA statement of methods. Veh Syst Dyn.
 2015;53(3):347-356. doi: 10.1080/00423114.2014.959027.
- [35] Zhou N, Lv Q, Yang Y, Zhang W. <TPL-PCRUN> Statement of methods. Veh Syst Dyn.
 2015;53(3):380-391. doi: 10.1080/00423114.2014.982136.
- [36] Ikeda K. Optimization of Overhead Contact Lines for Shinkansen Speed Increases, JR EAST Tech Rev. 2008;12:64-69.
- [37] Ambrosio J, Pombo J, Antunes P, Pereira M. Optimization of high-speed railway pantographs
 for improving pantograph-catenary contact, Theor Appl Mech Lett. 2013;3(1):013006. doi:
 10.1063/2.1301306.
- [38] Zhang W, Mei G, Wu X, Shen Z. Hybrid simulation of dynamics for the pantograph-catenary system. Veh Syst Dyn. 2002;38(6):393-414. doi:10.1076/vesd.38.6.393.8347
- [39] Zhang WH, Mei GM, Wu XJ, Chen LQ. A study on dynamic behaviour of pantographs by using hybrid simulation method. Proc Inst Mech Eng F, J Rail Rapid Transit. 2005;219(3):189-199. doi:10.1243/095440905X8880

- [40] Resta F, Facchinetti A, Collina A, Bucca G. On the use of a hardware in the loop set-up for pantograph dynamics evaluation. Veh Syst Dyn. 2008;46(S1):1039-1052.
 doi:10.1080/00423110802037891
- [41] Facchinetti A, Bruni S. Hardware-in-the-Loop Hybrid Simulation of Pantograph-Catenary Interaction. J Sound Vib. 2012;331:2783–2797. doi:10.1016/j.jsv.2012.01.033
- [42] Bruni S, Bucca G, Collina A, Facchinetti A. Numerical and hardware-in-the-loop tools for the design of very high speed pantograph-catenary systems. J Comput Nonlinear Dyn. 2012;7(4). doi:10.1115/1.4006834
- [43] Stoten DP, Yamaguchi T, Yamashita Y. Dynamically Substructured System Testing for Railway Vehicle Pantographs, Journal of Physics: Conference Series. 2016;744(1). doi:10.1088/1742-6596/744/1/012204
- [44] Schirrer A, Aschauer G, Talic E, Kozek M, Jakubek S. Catenary emulation for hardware-inthe-loop pantograph testing with a model predictive energy-conserving control algorithm. Mechatronics. 2017:41:17-28. doi: 10.1016/j.mechatronics.2016.11.002
- [45] Facchinetti A, Gasparetto L, Bruni S, Real-time catenary models for the hardware-in-the-loop simulation of the pantograph-catenary interaction. Veh Syst Dyn. 2013;51(4):499-516. doi: 10.1080/00423114.2012.748920
- [46] Gregori S, Tur M, Nadal E, Aguado JV, Fuenmayor FJ, Chinesta F. Fast simulation of the pantograph–catenary dynamic interaction. Finite Elements in Analysis and Design. 2017:;129:1-13. doi: 10.1016/j.finel.2017.01.007
- [47] Commission Regulation (EU) No 1301/2014 of 18 November 2014 on the technical specifications for interoperability relating to the 'energy' sub-system of the rail system in the Union.
- [48] European Committee for Electrotechnical Standardization (CENELEC). Railway applications
 Current collection systems Technical criteria for the interaction between pantograph and

overhead line (to achieve free access). Brussels: CENELEC;2012. Standard No. EN 50367:2012.

- [49] European Committee for Electrotechnical Standardization (CENELEC). Railway applications
 Fixed installations Electric traction overhead contact lines. Brussels: CENELEC;2009.
 Standard No. EN 50119:2009.
- [50] European Committee for Electrotechnical Standardization (CENELEC). Railway applications

 Current collection systems Requirements for and validation of measurements of the dynamic interaction between pantograph and overhead contact line. Brussels: CENELEC;2012. Standard No. EN 50317:2012.
- [51] Bucca G, Carnevale M, Comolli L, Facchinetti A, Cavaliere F, Pancari G, Brughera F, Noviello G. Assessment of current collection quality of ETR1000-V300 Zefiro pantograph: an innovative measurement set-up and test results. 11th World congress on railway research (WCRR 2016); 2016 May 29-Jun 2; Milan, Italy.
- [52] Masini P, Papi M, Puliatti G. Virtual acquisition system for experimentation in pantographcatenary interaction. 6th International Conference on Computer Aided Design, Manufacture and Operation in the Railway and Other Advanced Mass Transit Systems (COMPRAIL 98); 1998 Sept 2-4;Lisbon, Portugal. p.827-836.
- [53] Kolbe M, Baldauf W. Compact contact force measurement system online diagnosis of the overhead, line system with regular trains. 6th World congress on railway research (WCRR 2001); 2001 Nov 25-29; Cologne, Germany.
- [54] Paik JS, Park CK, Kim YG, Kim KH. A study on the measuring system of contact force of pantograph for the high-speed train. J Adv Mech Des Syst Manuf. 2012;6(1):140-148. doi: 10.1299/jamdsm.6.140.
- [55] Delfosse P, Sauvestre B. Measurement of contact pressure between pantograph and catenary. Fr Railw Rev. 1983;1(6):497-506.

- [56] Kiessling F, Puschmann R, Schmieder A, Schneider E. Contact Lines for Electric Railways:Planning, Design, Implementation, Maintenance. 2nd ed. Erlagen: Publicis CorporatePublishing; 2009.
- [57] Bocciolone, M., Comolli, L., Crosio, P., A novel design of a compact S-shaped balance with FBG sensors for the pantograph-catenary contact force measurement. Proceedings of SPIE -The International Society for Optical Engineering, 22nd International Conference on Optical Fiber Sensors; 2012 Oct 15-19; Beijing; China. art. no. 84214Z. doi: 10.1117/12.975146.
- [58] Bocciolone M, Bucca G, Collina A, Comolli L. Optical sensors for the analysis of pantograph-catenary interaction, Ing Ferrov. 2016;71(1):41-56
- [59] Schröder K, Ecke W, Kautz M, Willett S, Jenzer M, Bosselmann T. An approach to continuous on-site monitoring of contact forces in current collectors by a fiber optic sensing system. Opt Lasers Eng. 2013;51(2):172-179. doi: 10.1016/j.optlaseng.2012.08.007.
- [60] Bocciolone M, Bucca G, Collina A, Comolli L. Pantograph-catenary monitoring by means of fibre Bragg grating sensors: Results from tests in an underground line, Mech Syst Signal Process. 2013;41(1-2):226-238. doi: 10.1016/j.ymssp.2013.06.030.
- [61] Nåvik P, Rønnquist A, Stichel S. Variation in predicting pantograph-catenary interaction contact forces, numerical simulations and field measurements. Veh Syst Dyn. 2017;55(9):1265-1282. doi: 10.1080/00423114.2017.1308523.
- [62] Collina A, Lo Conte A, Carnevale M. Effect of collector deformable modes in pantographcatenary dynamic interaction. Proc Inst Mech Eng F, J Rail Rapid Transit. 2009;223(1):1-14. doi: 10.1243/09544097JRRT212.
- [63] Bruno O, Landi A, Papi M, Sani L. Phototube sensor for monitoring the quality of current collection on overhead electrified railways. Proc Inst Mech Eng F, J Rail Rapid Transit. 2001;215(3):231-241. doi: 10.1243/0954409011531549.

- [64] Barmada S, Landi A, Papi M, Sani L. Wavelet multiresolution analysis for monitoring the occurrence of arcing on overhead electrified railways, Proc Inst Mech Eng F, J Rail Rapid Transit. 2003;217(3):177-187. doi: 10.1243/095440903769012885.
- [65] Nåvik P, Rønnquist A, Stichel S. A wireless railway catenary structural monitoring system:Full-scale case study. Case Stud Struct Eng. 2016;6:22-30. doi: 10.1016/j.csse.2016.05.003.
- [66] Borgwartd H. Verschleiß Verhalten des Fahrdraht der Oberleitung. Elektr Bahnen. 1989.
- [67] Klapas D, Benson FA, Hackam R, Evison PR. Wear in simulated railway overhead current collection systems. Wear. 1988;126(2):167-190. doi: 10.1016/0043-1648(88)90136-6.
- [68] Kubo S, KatoK. Effect of arc discharge on the wear rate and wear mode transition of a copper-impregnated metallized carbon contact strip sliding against a copper disk. Tribol Int. 1999;32(7):367-378. doi: 10.1016/S0301-679X(99)00062-6.
- [69] Nagasawa, Kato K. Wear mechanism of copper alloy wire sliding against iron-base strip under electric current. Wear. 1998;216(2):179-183.
- [70] Chen GX, Yang HJ, Zhang WH, Wang X, Zhang SD, Zhou ZR. Experimental study on arc ablation occurring in a contact strip rubbing against a contact wire with electrical current. Tribol Int. 2013;61:88-94. doi: 10.1016/j.triboint.2012.11.020.
- [71] Bucca G, Collina A, Tanzi E. Experimental analysis of the influence of the electrical arc on the wear rate of contact strip and contact wire in a.c. system. Mech. Mach. Sci. 2017;47:449-456. doi: 10.1007/978-3-319-48375-7_48.
- [72] Ding T, Chen GX, Wang X, Zhu MH, Zhang WH, Zhou WX. Friction and wear behavior of pure carbon strip sliding against copper contact wire under AC passage at high speeds. Tribol Int. 2011;44(4):437-444. doi: 10.1016/j.triboint.2010.11.022.
- [73] He DH, Manory R, Grady N. Wear of railway contact wires against current collector materials. Wear. 1998;215(1-2):146-155.

- [74] Kubota Y, Nagasaka S, Miyauchi T, Yamashita C, Kakishima H. Sliding wear behavior of copper alloy impregnated C/C composites under an electrical current. Wear. 2013;302(1-2):1492-1498. doi: 10.1016/j.wear.2012.11.029.
- [75] Bruni S, Bucca G, Collina A, Facchinetti A, Melzi S. Pantograph-catenary dynamic interaction in the medium-high frequency range. Veh Syst Dyn. 2004;41(S1):697-706.
- [76] Bucca G, Collina A. A procedure for the wear prediction of collector strip and contact wire in pantograph–catenary system. Wear. 2009;266(1-2):46–59. doi: 10.1016/j.wear.2008.05.006.
- [77] Bucca G, Collina A. Electromechanical interaction between carbon-based pantograph strip and copper contact wire: A heuristic wear model. Tribol Int. 2015;92:47-56. doi: 10.1016/j.triboint.2015.05.019.
- [78] Massat JP, Tajan TN, Maitournam H, Balmes E, (2011) Fatigue Analysis of Catenary Contact Wires for High Speed Trains. 9th World Congress on Railway Research (WCRR 2011); 2011 May 22-26; Lille, France.
- [79] Zhen G, Kim Y, Haochuang L, Koo JM, Seok CS, Lee K, Kwon SY. Bending fatigue life evaluation of Cu-Mg alloy contact wire. Int J Precis Eng Manuf. 2014;15 (7):1331-1335. doi: 10.1007/s12541-014-0473-z.
- [80] Collina A, Fossati F, Papi M, Resta F. Impact of overhead line irregularity on current collection and diagnostics based on the measurement of pantograph dynamics. Proc Inst Mech Eng F, J Rail Rapid Transit. 2007;221(4):547–559. doi: 10.1243/09544097F02105.
- [81] Carnevale M, Collina A. Processing of collector acceleration data for condition-based monitoring of overhead lines. Proc Inst Mech Eng F, J Rail Rapid Transit. 2016;230(2):472-485. doi: 10.1177/0954409714545637.
- [82] Tanarro F, Fuerte V. OHMS Real-time analysis of the pantograph-catenary interaction to reduce maintenance costs. 5th IET Conference on Railway Condition Monitoring and Non-Destructive Testing (RCM 2011); 2011 Nov 29-30; Derby, UK. doi: 10.1049/cp.2011.0600.

- [83] Daadbin A, Rosinski J. Development, testing and implementation of the pantograph damage assessment system (PANDAS). WIT Trans Built Environ. 2010;114:573-578. doi: 10.2495/CR100531.
- [84] Landi A, Menconi L, Sani L. Hough transform and thermo-vision for monitoring pantographcatenary system. Proc Inst Mech Eng F, J Rail Rapid Transit. 2006;220(4):435-447. doi: 10.1243/0954409JRRT41.
- [85] Hamey LGC, Watkins T, Yen SWT. Pancam: In-service inspection of locomotive pantographs. Digital Image Computing Techniques and Applications: 9th Biennial Conference of the Australian Pattern Recognition Society (DICTA 2007); 2007 Dec 3-5; Glenelg, Australia. p. 493-499. doi: 10.1109/DICTA.2007.4426837.
- [86] Aydin I, Karakose M, Akin E. Anomaly detection using a modified kernel-based tracking in the pantograph-catenary system. Expert Sys Appl. 2015;42(2):938-948. doi: 10.1016/j.eswa.2014.08.026.
- [87] Bocciolone M, Resta F, Rocchi D, Tosi A, Collina A. Pantograph aerodynamic effects on the pantograph–catenary interaction. Veh Syst Dyn. 2006;44(S1):560-570. doi: 10.1080/00423110600875484.
- [88] Pombo J, Ambrosio J, Pereira M, Rauter F, Collina A, Facchinetti A. Influence of the aerodynamic forces on the pantograph-catenary system for high-speed trains. Veh Syst Dyn. 2009;47(11):1327-1347.
- [89] Carnevale M, Facchinetti A, Robustelli F, Rocchi D. Unsteady aerodynamic forces on a railway pantograph and their influence on pantograph-catenary dynamic interaction. 25th IAVSD International Symposium on Dynamics of Vehicles on Roads and Tracks; 2017 Aug 14-18; Rockhampton, Australia.
- [90] Lee Y, Paik J, Kwak M, Yoo J, Kim KH, Lee D, Kwon H. Analysis and comparison of experimental results both of wind tunnel test and running test for HEMU-400X pantograph.
 2012 Joint Rail Conference (JRC 2012); 2012 April 17-19, Philadelphia PA, USA. p.169-172.

- [91] Lv Q, Li R, Wang S, Zhou N, Zhang W. The Effect of the Working Height of Pantographs on Pantograph-Catenary Dynamic Performance. Civil-Comp Proc. 2014;104.
- [92] Carnevale M, Facchinetti A, Maggiori L, Rocchi D. Computational fluid dynamics as a means of assessing the influence of aerodynamic forces on the mean contact force acting on a pantograph. Proc Inst Mech Eng F, J Rail Rapid Transit. 2015;230(7):1698–1713. doi: 10.1177/0954409715606748.
- [93] Carnevale M, Facchinetti A, Rocchi D. Procedure to assess the role of railway pantograph components in generating the aerodynamic uplift. J Wind Eng Ind Aerodyn. 2017;160:16-29. doi: 10.1016/j.jweia.2016.11.003.
- [94] Li R, Zhang W, Ning Z, Liu B, Zou D, Liu W. Influence of a high-speed train passing through a tunnel on pantograph aerodynamics and pantograph-catenary interaction. Proc Inst Mech Eng F, J Rail Rapid Transit. 2015;231(2):198-210. doi: 10.1177/0954409715626743.
- [95] Ikeda M, Manabe K. Development of low noise pantograph with passive lift suppression mechanism of panhead. Q Rep of RTRI. 2000;41(4):177-181.
- [96] Carnevale M, Facchinetti A, Rocchi D. Assessing aerodynamic effects on a railway pantograph by means of computational fluid dynamics. Civil-Comp Proc. 2016;110.
- [97] Lee J, Cho W. Prediction of low-speed aerodynamic load and aeroacoustic noise around simplified panhead section model. Proc Inst Mech Eng F, J Rail Rapid Transit, 2008;222(4):423–431. doi: 10.1243/09544097JRRT152.
- [98] Suzuki M, Ikeda M, Koyama T. Flow control on pantograph with air intake and outlet. Q Rep of RTRI. 2007;48(4):236–239. doi: 10.2219/rtriqr.48.236.
- [99] Lee Y, Rho J, Kwak M, Lee J, Kim K, Lee D. Aerodynamic characteristics of high speed train pantograph with the optimized panhead shape. 7th IASME/WSEAS International Conference on Fluid Mechanics and Aerodynamics (FMA '09); 2009 Aug 20-22; Moscow, Russian Federation. p. 84-88.

- [100] Tuissi A, Bassani P, Casati R, Bocciolone M, Collina A, Carnevale M, Lo Conte A, Previtali B. Application of SMA composites in the collectors of the railway pantograph for the Italian high-speed train. J Mater Eng Perform. 2009;18(5-6):612-619. doi: 10.1007/s11665-009-9453-3.
- [101] Ikeda M, Mitsumoji T. Numerical estimation of aerodynamic interference between panhead and articulated frame. Q Rep of RTRI. 2009;50(4):227–232. doi: 10.2219/rtriqr.50.227.
- [102] Takaishi T, Ikeda M. Experimental method for wind tunnel tests to simulate turbulent flow on the roof of high-speed trains. Q Rep of RTRI. 2012;53(3):167-172. doi: 10.2219/rtriqr.53.167.
- [103] Ikeda M, Mitsumoji T. Evaluation Method of Low-Frequency Aeroacoustic Noise Source Structure Generated by Shinkansen Pantograph. Q Rep of RTRI. 2008;49(3):184-190. doi: 10.2219/rtriqr.49.184.
- [104] Yao S, Guo D, Yang G. The influence of pantograph aerodynamic characteristics caused by its shroud. 1st International workshop on high-speed and intercity railways (IWHIR 2011); 2011 Jul19-22; Shenzhen, China. p. 41-52. doi: 10.1007/978-3-642-27963-8_5.
- [105] Luo J, Luo J, Yang Z, Chen Y, Gao Z. Numerical research on aerodynamic characteristic optimization of pantograph fixing place on high speed train. 2nd Conference on Power Electronics and Intelligent Transportation System (PEITS 2009); 2009 Dec 19-20; Shenzhen, China. p. 94-97. doi: 10.1109/PEITS.2009.5406957.
- [106] Zhang L, Zhang J, Li T, Zhang W. Influence of pantograph fixing position on aerodynamic characteristics of high-speed trains. J Mod Transp. 2017;25 (1)34-39. doi: 10.1007/s40534-017-0125-y.
- [107] Gregoire R, Collina A, Resta F, Rocchi D. Some considerations on the aerodynamics of high speed pantograph: CFD and wind tunnel results. BBAA VI International Colloquium on: Bluff Bodies Aerodynamics & Applications; 2008 Jul -20-24; Milan, Italy.
- [108] Wann LM. Improvement of a pantograph for high speed train [master's thesis]. Cambridge (MA): Massachusetts Institute of Technology; 1980.

- [109] Vinayagalingam T. Computer evaluations of controlled pantographs for current collection from simple catenary overhead equipment at high speed. J Dyn Syst Meas Control Trans ASME. 1983;105(4):287-294. Doi: 10.1115/1.3140673.
- [110] Baldauf W, Blaschko R, Behr W, Heine C, Kolbe M. Development of an actively controlled, acoustically optimised single arm pantograph. 6th World congress on railway research (WCRR 2001); 2001 Nov 25-29; Cologne, Germany.
- [111] Allotta B, Papi M, Pugi L, Toni P, Violi G. Experimental campaign on a servo-actuated pantograph. IEEE/ASME international conference on advanced intelligent mechatronics (AIM2001); 2001 Jul 8-12; Como, Italy. P. 237-242. doi: 10.1109/AIM.2001.936460.
- [112] Facchinetti A, Mauri M. Hardware-in-the-loop overhead line emulator for active pantograph testing. IEEE Trans Ind Electron. 2009;56(10):4071-4078. doi: 10.1109/TIE.2009.2023632.
- [113] Allotta B, Pugi L, Bartolini F. Design and experimental results of an active suspension system for a high-speed pantograph. IEEE ASME Trans Mechatron . 2008:13(5);548-557. doi: 10.1109/TMECH.2008.2002145.
- [114] Baldauf W, Kolbe M, Krötz W. [Closed-loop controlled pantograph for high-speed applications], eb Elektr Bahnen. 103(4), 225-230, 2005. German.
- [115] Cathala J, ForteP, inventors; Faiveley Transport, assignee. Device for adjusting the bearing force of a pantograph o on a catenary wire and process relating thereto. United States patent US 5115405. 1992 May 19.
- [116] Bruni S, Carnevale M, Facchinetti A. Active control of the pantograph articulated frame to reduce contact force fluctuations in the low frequency range. 13th Mini Conference on Vehicle System Dynamics, Identification and Anomalies (VSDIA 2012); 2012 Nov 5-7; Budapest, Hungary. p. 237-244.
- [117] Bucca G, Carnevale M, Collina A, Facchinetti A. An active control strategy for multiple pantograph collection. 23rd International Symposium on Dynamics of Vehicles on Roads and Tracks (IAVSD2013); 2013 Aug 19-23; Qingdao, China.

- [118] Galeotti G, Galanti M, Magrini S, Toni P. Servo actuated railway pantograph for high-speed running with constant contact force. Proc Inst Mech Eng F, J Rail Rapid Transit, 1993:207(1):37-49. doi: 10.1243/PIME_PROC_1993_207_225_02.
- [119] Althammer KA, Baldauf W. Considerations on high performance pantographs. 4th World Congress on Railway Research (WCRR'99); 1999 Oct 19-23; Tokio, Japan.
- [120] Collina A, Facchinetti A, Fossati F, Resta F. An application of active control to the collector of an high-speed pantograph: simulation and laboratory tests. 44th IEEE Conference on Decision and Control, and the European Control Conference, CDC-ECC '05; 2005 Dec 12-15; Seville, Spain. p. 4602-4069. doi: 10.1109/CDC.2005.1582888.
- [121] Balestrino A, Bruno O, Landi A, Sani L. Innovative solutions for overhead catenarypantograph system: Wire actuated control and observed contact force. Veh Syst Dyn. 2000;33(2):69-89.
- [122] Pisano A, Usai E. Contact force regulation in wire-actuated pantographs via variable structure control and frequency-domain techniques. Int J Control. 2008;81(11):1747-1762. Doi: 10.1080/00207170701874473.
- [123] Collina A, Facchinetti A, Resta F. A feasibility study of an aerodynamic control for a high speed pantograph. IEEE/ASME Internation Conference on Advanced Intelligent Mechatronics, AIM 2007; 2007 Sept 4-7, Zurich, Switzerland. art. no. 4412560. doi: 10.1109/AIM.2007.4412560.
- [124] Sanchez-Rebollo C, Jimenez-Octavio JR, Carnicero A. Active control strategy on a catenarypantograph validated model. Veh Syst Dyn. 2013;51(4):554-569. doi: 10.1080/00423114.2013.764455.
- [125] Song Y, Ouyang H, Liu Z, Mei G, Wang H, Lu X. Active control of contact force for highspeed railway pantograph-catenary based on multi-body pantograph model. Mech Mach Theory . 2017;115:35-59. doi: 10.1016/j.mechmachtheory.2017.04.014.

- [126] Gaing ZL, Chang RF. Optimal PID controller for high-speed rail pantograph system with notch filter. 2009 IEEE Region 10 Annual International Conference, TENCON 2009; 2009 Nov 23-26; Singapore. art. no. 5396061. doi: 10.1109/TENCON.2009.5396061.
- [127] Rusu-Anghel S, Miklos C, Averseng J, Tirian GO. Control system for catenary-pantograph dynamic interaction force. International Joint Conference on Computational Cybernetics and Technical Informatics (ICCC-CONTI 2010); 2010 May 27-29; Timisoara, Romania. p. 181-186. doi: 10.1109/ICCCYB.2010.5491302.
- [128] Walters S. Simulation of fuzzy control applied to a railway pantograph-catenary system. 14th International Conference on Knowledge-Based and Intelligent Information and Engineering Systems (KES 2010); 2010 Sept 9-10; Cardiff; UK. p.322–330. doi: 10.1007/978-3-642-15390-7_33.
- [129] Diana G, Fossati F, Resta F. High speed railway: pantograph active control and overhead lines diagnostics. Veh Syst Dyn. 1998;30(1):69-84. doi: 10.1080/00423119808969436.
- [130] Carillo D, Collina A, Facchinetti A, Fossati F, Papi M, Resta F. Application of EKF to the estimation of contact forces in pantograph-catenary system. 7th World congress on railway research (WCRR 2003); 2003 Sept 28-Oct 1; Edinburgh, UK.
- [131] Pisano A, Usai E. Contact force estimation and regulation in active pantographs: An algebraic observability approach. 46th IEEE Conference on Decision and Control; 2007 Dec 12-14; New Orleans LA; US. p. 4341–4346. doi: 10.1109/CDC.2007.4434612.
- [132] Ko MT, Yokoyama M, Yamashita Y, Kobayashi S, Usuda T. Contact force control of an active pantograph for high speed trains. J Phys Conf Ser. 2016;744(1):012151. doi: 10.1088/1742-6596/744/1/012151.
- [133] O'Connor D, Eppinger S, Seering W, Wormley D. Active control of a high-speed pantograph.J Dyn Syst Meas Control Trans ASME. 1997;119(1):1-4. doi: 10.1115/1.2801209.
- [134] Wu TX, Brennan MJ. Active vibration control of a railway pantograph. Proc Inst Mech Eng F, J Rail Rapid Transit. 2005;211(2):117-130. doi: 10.1243/0954409971530969.

- [135] Shudong W, Jingbo G, Guosheng G. Research of the active control for high-speed train pantograph. 2008 IEEE Conference on Cybernetics and Intelligent Systems (CIS 2008); 2008 Sept 21-24; Chengdu, China. p. 749-753. doi: 10.1109/ICCIS.2008.4670754.
- [136] Pappalardo CM, Patel MD, Tinsley B, Shabana AA. Contact force control in multibody pantograph/catenary systems. Proc Inst Mech Eng Part K, J Multi-body Dyn. 2016;230(4):307-328. doi: 10.1177/1464419315604756.
- [137] Song Y, Liu Z, Ouyang H, Wang H, Lu X. Sliding mode control with PD sliding surface for high-speed railway pantograph-catenary contact force under strong stochastic wind field. Shock Vib. 2017:art. no. 4895321. doi: 10.1155/2017/4895321.

Table 1: Caption

Figure 1: Caption